Author’s response to reviews

Title: Colorectal carcinoma in the course of inflammatory bowel diseases

Authors:

Andrzej Hnatyszyn (andrzej.hnatyszyn@gmail.com)
Szymon Hryhorowicz (szymon.hryhorowicz@gmail.com)
Marta Kaczmarek-Ryś (m.kaczmarekrys@gmail.com)
Emilia Lis (emilialis31@gmail.com)
Ryszard Slomski (slomski@au.poznan.pl)
Rodney Scott (Rodney.Scott@newcastle.edu.au)
Andrzej Plawski (andp@man.poznan.pl)

Version: 1 Date: 21 Jun 2019

Author’s response to reviews:

We are grateful for the valuable instructions and critical comments.

Following the advice of the first Reviewer, we decided to replace all terms described in full by abbreviations e.g. we changed “colorectal cancer” into CRC, “ulcerative colitis” into UC “Crohn's disease” into CD and “Familial adenomatous polyposis” into FAP.

According to the second Reviewer comments, we admit that we had made a mistake using the word “indisputable”, which was replaced by “unquestionable”. We believe that this word is more proper in a scientific journal, also the final correction of the manuscript was made by the native speaker (professor of human genetics).

The manuscript version with tracked changes is attached.